Author's response to reviews

Title: Economic valuation of informal care on cerebrovascular accident survivors

Authors:

Juan Oliva-Moreno (juan.olivamoreno@uclm.es)
Isaac Aranda-Reneo (isaac.aranda@uclm.es)
Cristina Vilaplana (cvilaplana@um.es)
Almudena Gonzalez-Dominguez (almudena.gonzalez@imw.es)
Álvaro Hidalgo-Vega (alvaro.hidalgo@uclm.es)

Version: 9 Date: 11 September 2013

Author's response to reviews: see over
Dear Editor,

The authors appreciate the comments received by the reviewer.

The reviewer comments are answered below:

#1
The authors have included in page 8 the following paragraph in order to explain the tasks that are included in “social services home care” to people from other countries.

“In Spain, the social services home care are made up of a set of initiatives that are performed in the home of the dependent person in order to cater for his/her everyday needs. It includes: (a) services related to attending to domestic or home needs (cleaning, ironing, washing, cooking and others); (and b) services related to personal care, in performing the activities of daily living (bathing, dressing, helping to walk, etc.)”

#2
The referee suggested changing in Discussion section paragraph 1 “solid” to “robust”. Now it appears “robust” instead of “solid”.

#3
Finally one of the authors (Cristina Vilaplana) wants to acknowledge the help provided by project ECO2011-30323-C03-02 and ECO2011-2850 from the Spanish Ministry of Science and Innovation. So we have included this information in the acknowledgments section.